Author's response to reviews

Title: FUMEPOC Early detection of chronic obstructive pulmonary disease in smokers

Authors:

Vicente Gil-Guillén (atencion.primaria@umh.es)
Domingo Orozco-Beltrán (dorozcobeltran@gmail.com)
Concepcion V Carratala Munuera (atencion.primaria@umh.es)
Carlos Plaza-Sirvent (biokalos@hotmail.com)
Patricia Lorca-Amorrich (patri_lorca@hotmail.com)
Adriana López-Pineda (adriannalp@hotmail.com)
María P Vela-Troncoso (aki_pilar2002@hotmail.com)
Juan J Soler (jjsoler@telefonica.net)
Manuel Yarza-Cañellas (yarza_man@gva.es)
Antonio Fernández (tonifdz@yahoo.es)
Luis Rosado-Bretón (rosado_lui@gva.es)

Version: 5 Date: 13 May 2011

Author's response to reviews:

Dear editorial board of BMC Public Health,

Please find enclosed the revised manuscript: “FUMEPOC. Early detection of chronic obstructive pulmonary disease in smokers”, by Vicente Gil-Guillen, et al.

We have made all the changes you required us (major revisions and minor revisions. Moreover, we have added the names of the members of a clinical group, which will be responsible for carrying out the study, to the author list on the title page as you suggest us.

Thanks for your suggestions which will improve the manuscript.

Sincerely yours,

Vicente Gil-Guillen on behalf of the authors.